Author's response to reviews

Title: Monocytes and neutrophils expressing myeloperoxidase occur in fibrous caps and thrombi in unstable coronary plaques

Authors:

Fabio Tavora (ftavora@gmail.com)
Mary Ripple (ripplem@ocmemd.org)
Ling Li (ling001@aol.com)
Allen Burke (allen.burke@gmail.com)

Version: 4 Date: 1 June 2009

Author's response to reviews:

Dear Editor-in-chief,

Enclosed please find the revised manuscript entitled “Monocytes and neutrophils expressing myeloperoxidase occur in fibrous caps and thrombi in unstable coronary plaques”.

In the revised manuscript, all formatting changes are done as instructed by the last correspondence. In addition, figure references were introduced in the text. Figure 5 contained redundant information and was deleted from the final manuscript.

Sincerely,

Allen Burke, MD
Associate Professor
University of Maryland School of Medicine
Department of Pathology
22 S Greene St Room NBW64
Baltimore, MD 21201
Phone: 410 3285525
Fax: 4103285508